Author’s response to reviews

Title: Effectiveness of Evidence-Based Medicine on Knowledge, Attitudes, and Practices of Family Planning Providers: A Randomized Experiment in Jordan

Authors:

Marianne El-Khoury (Marianne_ElKhoury@abtassoc.com)
Rebecca Thornton (rebeccat@illinois.edu)
Minki Chatterji (Minki_Chatterji@abtassoc.com)
Soonie Choi (soonie_choi@abtassoc.com)

Version: 4  Date: 16 September 2015

Author's response to reviews: see over
September 15, 2015

To the editors of BMC Health Services Research,

We made the following changes to the manuscript to conform to the journal style:

- Formatted the abstract sections according to the Instruction for Authors.
- Deleted the Funding section. The main funding source (U.S. Agency for International Development) and funding from Abt Associates for manuscript development were already mentioned in Acknowledgements. Added the Abt Associates funding in the competing interests section.
- Revised the title page of the manuscript.
- Removed unnecessary capitalization from section and subsection headings and from table and figure titles.
- Placed the reference numbers in square brackets before the full stop, as instructed.
- Changed the font and line spacing according to the template provided.
- Updated two references.
- Added visible black lines to table borders.
- Removed line numbering.

We uploaded the revised manuscript to the website, as instructed.

Please let us know if you have any additional questions. We look forward to hearing from you.

Best regards,

Marianne El-Khoury
Marianne_Elkhoury@abtassoc.com
Phone: 301.347.5227